Author's response to reviews

Title: Incidence and Risk Factors of infection in oral cancer patients undergoing different treatments protocols

Authors:

Manju Panghal (manju_hooda2007@rediffmail.com)
Vivek Kaushal (vkaushal@gmail.com)
Sangeeta Kadayan (sangeeta.kadayan@rediffmail.com)
Jaya Parkash Yadav (yadav1964@rediffmail.com)

Version: 3 Date: 11 June 2012

Author's response to reviews:

Ms Wella Valenzuela

The BioMed Central Editorial Team

Subject: Submission of Revised Ms entitled “Incidence and Risk Factors of infection in oral cancer patients undergoing different treatments protocols”.

Madam

Thank you for accepting the Ms. I am again submitting the revised Ms in view of minor correction raised by respected reviewer in Table 5 and correction in language throughout the Ms. The necessary corrections have been shown in yellow color. I hope the revised Ms fulfil the journal criteria.

Reply to Reviewer's report

Reviewer: Velia Ramirez Amador

Reviewer's report:

The authors have answered all the suggestions made.

It is only recommended to revise Table 5, for spelling mistakes. Also, to revise the spelling and grammar of the text for minor mistakes.

Answer: The Table 5 has been revised by correcting the spelling mistakes. The spellings and grammar throughout the Ms has been corrected and shown in yellow color.

Submitted for your kind consideration.

With best regards

J.P. Yadav